Author’s response to reviews

Title: Development of an interactive web-based tool to conduct and interrogate meta-analysis of diagnostic test accuracy studies: MetaDTA

Authors:

Suzanne Freeman (suzanne.freeman@leicester.ac.uk)

Clareece Kerby (clareece.kerby@nhs.net)

Amit Patel (ap666@le.ac.uk)

Nicola Cooper (njc21@le.ac.uk)

Terry Quinn (terry.quinn@glasgow.ac.uk)

Alex Sutton (ajs22@le.ac.uk)

Version: 3 Date: 20 Mar 2019

Author's response to reviews:

Editor comments:

1. To meet the submission guidelines of the journal please restructure your abstract. Your abstract should include the following headings: Background, Methods, Results, Conclusions, and should be no longer than 350 words. Please ensure you apply this format to the abstract in the main text and the submission system also.

Response: The abstract contains the headings Background, Methods, Results, Conclusions and in both Microsoft Word and the online submission system the length of the abstract is 349 words.

2. To meet the submission guidelines of the journal please include the following headings within your manuscript: Background, Implementation, Results, Discussion (if appropriate), Conclusion, Availability and Requirements, Abbreviations, Declarations.
Response: The manuscript has been updated to include the following headings: Background, Implementation, Results, Discussion, Conclusion, Availability and Requirements, Abbreviations, Declarations.